Author's response to reviews

Title: Organising health care services for people with an acquired brain injury: An overview of systematic reviews and randomised controlled trials

Authors:

Kate Laver (Kate.Laver@health.sa.gov.au)
Natasha A Lannin (N.Lannin@latrobe.edu.au)
Peter Bragge (peter.bragge@monash.edu)
Peter Hunter (Peter.Hunter@cgmca.org.au)
Anne E Holland (A.Holland@latrobe.edu.au)
Emma Tavender (emma.tavender@monash.edu)
Denise O'Connor (denise.oconnor@monash.edu)
Fary Khan (fary.khan@mh.org.au)
Robert Teasell (robert.teasell@sjhc.london.on.ca)
Russell Gruen (russell.gruen@monash.edu)

Version: 3 Date: 22 July 2014

Author's response to reviews:

Dear Reviewers,

Thank you for your comments and suggestions. A summary of revisions made is attached to this submission

Kind regards